Author's response to reviews

Title: Ethnic differences translate to inadequacy of high-risk screening for gestational diabetes mellitus in an Asian population: a cohort study

Authors:

Yap-Seng Chong (obgcys@nus.edu.sg)
Shirong Cai (obgcais@nus.edu.sg)
Harvard Lin (harvard_zj_lin@nuhs.edu.sg)
Shu E Soh (ephsse@nus.edu.sg)
Yung-Seng Lee (paeleeys@nus.edu.sg)
Melvin Khee-Shing Leow (melvin_leow@sics.a-star.edu.sg)
Yiong-Huak Chan (medcyh@nus.edu.sg)
Li Chen (chen_li@sics.a-star.edu.sg)
Joanna D Holbrook (joanna_holbrook@sics.a-star.edu.sg)
Kok-Hian Tan (Tan.Kok.Hian@kkh.com.sg)
Victor Samuel Rajadurai (victor.samuel@kkh.com.sg)
George Seow-Heong Yeo (Dr.George.SH.Yeo@gmail.com)
Michael S Kramer (michael.kramer@mcgill.ca)
Seang-Mei Saw (ephssm@nus.edu.sg)
Peter D Gluckman (pd.gluckman@auckland.ac.nz)
Keith M Godfrey (kmg@mrc.soton.ac.uk)
Kenneth Kwek (Kenneth.Kwek.YC@kkh.com.sg)

Version: 2 Date: 6 August 2014

Author's response to reviews:

Please see attached cover letter for point-by-point reply to reviewers' comments and suggestions.